Author’s response to reviews

Title: Bioelectronic DNA Detection of Human Papillomaviruses Using eSensor: A Model System for Detection of Multiple Pathogens

Authors:

Dr Suzanne D Vernon (svernon@cdc.gov)
Daniel H Farkas (dfarkas@bcm.tmc.edu)
Elizabeth R Unger (eru0@cdc.gov)
Vivian Chan (vvyc@yahoo.com)
Donna L Miller (dlm5@cdc.gov)
Yin-Peng Chen (yin.peng@motorola.com)
Gary F Blackburn (gary.blackburn@motorola.com)
Dr William C Reeves (wcr1@cdc.gov)

Version: 3 Date: 15 Jun 2003

I have updated an email address of one of the coauthors (B Chen) and added EDRN (Early Detection Research Network, NCI) in the acknowledgements.